Author's response to reviews

Title: Therapeutic Efficacy of Traditional Chinese Medicine, Shen-Mai San in Cancer Patient Undergoing Chemotherapy or Radiotherapy: study protocol for a Randomized, Double-blind, Placebo-controlled Trial

Authors:

Lun-Chien Lo (126478@cch.org.tw)
Chia-Yun Chen (137877@cch.org.tw)
Shou-tung Chen (1886@cch.org.tw)
Hung-Chang Chen (54464@cch.org.tw)
Tsung-Chieh Lee (129289@cch.org.tw)
Cheng-Shyong Chang (15120@cch.org.tw)

Version: 4 Date: 13 November 2012

Author's response to reviews:

Dear Editor,

Thank you for your useful comments and suggestions on the language and structure of our manuscript. Based the comments from the referees, we have changed title format and making correction as mentioned. Besides, We also modified the manuscript as the instructions.